Author's response to reviews

Title: Potential risk factors for diabetic neuropathy: a case control study

Authors:

Fargol Booya (emrc@tums.sina.ac.ir)
Fatemeh Bandarian (emrc@tums.sina.ac.ir)
Bagher Larijani (emrc@tums.sina.ac.ir)
Mohammad Pajouhi (emrc@tums.sina.ac.ir)
Mahdi Nooraei (emrc@tums.sina.ac.ir)
Jamshid Lotfi (emrc@tums.sina.ac.ir)

Version: 5 Date: 26 November 2005

Author's response to reviews:

Re: "Potential risk factors of diabetic neuropathy: a cross sectional study" (MS#1315091906736316)

Dear Editor
The requested formatting requirements was checked again throughout the text and modified considering manuscript formatting checklist and it was finalized. Also the new version of figures with EPS format was provided.

Best Regards,
Bagher Larijani